

Recommendations for an Interconnected Arctic

**Telecommunications Infrastructure Working Group** 



### Table of Contents

# ▶ AEC Telecommunications Infrastructure Working Groups ....31

principles of partnership, collaboration, innovation and peace. Being raised in rural Alaska, I have a deep understanding for the importance of connectivity and the challenges that come with a lack of reliable communications. Expanding broadband access and adoption will be vital for the economic, social and political growth of local Arctic communities. It is my hope that these recommendations add value to the ongoing discussion of broadband deployment in the Arctic, and serve as a tool for policy makers, investors, researchers and communities to come together for sustainable polar growth.

Asweeney.

# Message from AEC Telecommunications Infrastructure Working Group chair, Robert McDowell



The recommendations provided in this report are the result of a true collaborative effort among the business community within the eight Arctic states. Together, local Arctic residents and expert broadband advisors have combined their knowledge to establish a comprehensive strategy for the deployment and adoption of broadband in the far north - a first of its kind. To the members of the Telecommunications Infrastructure Working group, I thank you for your unwavering commitment and invaluable input contributed throughout this process.

May this document mark the beginning of increased broadband access in the Arctic, and encourage continued work in the field of broadband deployment throughout the circumpolar north.

Routin morrell





## Message from AEC chair, Tara Sweeney

I am incredibly proud of the hard work and dedication demonstrated by the members of the Telecommunications Infrastructure Working group. The pan-Arctic engagement evident throughout this document exhibits the strong commitment of the Arctic business community to support the Arctic Economic Council's four core



With its beautiful scenery, harsh climate, significant indigenous population and increasing geopolitical importance, the Arctic is unique. The challenges posed by the Arctic to broadband deployment and adoption are also unlike any on the planet. With this in mind, the dual purposes of the *Recommendations for an Interconnected Arctic* are to take stock of the current state of broadband in the Arctic, and to make recommendations as to how to facilitate broadband deployment and adoption.

This document begins by exploring how each Arctic state (Canada, Denmark, Finland, Iceland, Norway, Russia, Sweden and United States of America) defines broadband and the goals each has established for broadband deployment. Typically, Arctic states define broadband in terms of data transmission speed or have established transmission speed goals. Several Arctic states have established speed goals of 100 Mbit/s by the year 2020.

Next, this document lays out some of the overarching societal benefits of broadband. From the economy and healthcare to scientific research and public safety – broadband has the potential to positively affect nearly every sector of society. It facilitates and enhances our daily lives in ways once unimaginable. Indeed, broadband has the power to transform society and enable new and more robust ways of interacting with one another.

At the same time, the Arctic poses extreme challenges. The region is characterized by a dispersed population, harsh climate and lower rates of broadband adoption than non-Arctic regions. This document discusses these and other challenges that must be considered and surmounted in order to expand broadband in the Arctic.

These *Recommendations for an Interconnected Arctic* note that a comprehensive strategy for the Arctic requires a multi-pronged approach focused on ways to promote deployment while encouraging broadband demand.

Understanding current broadband usage by country is a necessary first step in realizing the ultimate goal of making the technology universally available in the region. As such, this document looks at current broadband deployment in each Arctic state. All but one Arctic state have overall broadband coverage rates of 90 percent or higher.

The following assessment then provides an overview of some of the existing technologies that may facilitate Arctic broadband deployment. Technologies such as optical fiber, mobile wireless and satellite each have positive and negative aspects that should be considered and evaluated. No single technology is appropriate for all situations. Rather, broadband deployment in the Arctic likely requires a combination of various technologies.

Next, funding options are explored. Private investment may be difficult to attract to Arctic broadband projects given the high cost of deployment and often small clusters of customers. Nevertheless, expanded oil and gas

exploration and other business activities may facilitate improved access to private finance. This document identifies numerous policy initiatives, such as those that adopt market mechanisms to promote more efficient spectrum use that may attract private investment in broadband. Public-private partnerships (PPPs) can also be an effective tool to promote delivery of broadband services to Arctic populations, and this document enumerates concrete steps governments and regulators can take to foster PPPs. Lastly, government support of Arctic broadband deployment is an option and can take a number of forms, including subsidies, tax incentives and the creation of universal service funds.

Past, current and proposed broadband deployment projects are the next topic addressed. Among these are the Quintillion project that will build and operate a privately-funded, 1,850 km long subsea fiber optic system that will initially run along the north and west coast of Alaska; and a project funded by Nornickel, a privately-owned Russian mining and metallurgical company that will construct a 957 km long fiber optic cable laid through the Yamal-Nenets Autonomous Region and Krasnoyarsk Territory. Numerous other projects are also described.

This document concludes by identifying goals and recommendations for Arctic broadband deployment and adoption. Included among these are examples of potential legal, regulatory, and business practice reforms and strategies that the private sector, NGOs and local, indigenous, provincial, and national governments could undertake to facilitate deployment. For example, governments could modernize tax structures to incentivize private investment in broadband and make more spectrum available.

As to adoption, this document encourages the private and public sectors to work together to address barriers to entry, including cost, and to further promote digital literacy.





# I. Introduction

The interconnectivity of people, communities, governments, businesses and beyond is one of the hallmarks of these opening decades of the 21st century. Reliable broadband is necessary to promote and advance interconnectivity, which in turn facilitates improvements in national economies, education, health, and many other sectors of society. Despite these benefits, broadband deployment and adoption across the globe have not been uniform. One region in danger of being left behind is the Arctic.

Recognizing this, private sector representatives from each nation state with territory in the Arctic – Canada, Denmark, Finland, Iceland, Norway, Russia, Sweden and the United States of America – have come together to take stock of the current state of broadband in the Arctic and consider how to promote greater interconnectivity. This document is the result of this unprecedented collaboration. However, it is not the end point. Given the constant progression of technology generally, and broadband specifically, strategies designed to advance broadband deployment and adoption must necessarily be iterative.

This document proceeds in five substantive sections. Section II provides a snapshot of broadband in the Arctic, including the current state of broadband coverage and adoption, challenges to expanding broadband and technologies available to facilitate broadband deployment. Section III summarizes broadband usage in each Arctic state and the technologies available to effectuate broadband connectivity. Section IV discusses the funding options available for the further development of broadband infrastructure. Section V provides an overview of various broadband projects in and adjacent to the Arctic. Section VI lays out goals and recommendations for a regional strategy to promote broadband deployment and adoption throughout the Arctic.



### II. Key Issues

#### **A. BROADBAND PERSPECTIVES**

In general terms, broadband refers to telecommunications services that provide multiple channels of data over a single communications medium, typically using some form of frequency or wave division multiplexing.<sup>1</sup> More specific definitions vary, with no international consensus as to how broadband should be defined.<sup>2</sup> As discussed below, the majority of Arctic states define broadband in terms of data transmission speed or have set transmission speed goals.<sup>3</sup>

1. Country Definitions/Goals

#### Canada

Two regulatory agencies – the Innovation, Science and Economic Development (ISED) agency and the Canadian Radio-Television and Telecommunications Commission (CRTC) – have jurisdiction over broadband-related issues in Canada. The ISED defines broadband as an "Internet service that is always on (as opposed to dial-up, where a connection must be made each time) and offers higher speeds than dial-up service."<sup>4</sup> The agency further specifies that "In Canada, broadband service refers to download speeds of 1.5 [Mbit/s] or greater."<sup>5</sup> The CRTC takes a slightly different view. In 2011, the CRTC set a broadband speed goal of 5 Mbit/s download and 1 Mbit/s upload for all Canadians.<sup>6</sup>

#### Denmark

In 2013, the Kingdom of Denmark's Ministry of Business and Growth established a broadband speed goal of at least 100 Mbit/s for all households and businesses by 2020.<sup>7</sup>

#### Finland

The Republic of Finland considers broadband access to be a basic legal right.<sup>8</sup> In 2008, Finland promulgated a national broadband plan that called for construction of fiber middle-mile networks such that at least one fiber access point is located no further than 2 km from virtually all permanent residences and businesses.<sup>9</sup> These networks are intended to cover 99 percent of the Finnish population.<sup>10</sup> In November 2015, Finland's Ministry of Transport and Communications (MTC) established a minimum speed rate of 2 Mbit/s for universal broadband service with a goal of increasing the speed to 10 Mbit/s by 2021.<sup>11</sup> In 2016, the MTC promulgated a broadband implementation plan designed to facilitate build-out of the fiber networks construction by increasing market-driven supply and demand without distorting competition and while utilizing a technology-neutral approach.<sup>13</sup>





Additionally, telecom operators defined as "universal service providers" must be able to provide every permanent residence and business office with access to a reasonably priced and high-quality connection with a downstream rate of at least 1 Mbit/s.<sup>14</sup>

#### Iceland

The Republic of Iceland broadly defines broadband as "high-speed Internet connections over fibre optic cables."<sup>15</sup> Currently, Internet service providers in Iceland provide fiber speeds ranging from 40 Mbit/s to 500 Mbit/s.<sup>16</sup>

#### Norway

The Kingdom of Norway's National Communications Authority ("NCOM," previously the National Post and Telecommunications Authority) defines broadband as having a minimum service capacity greater than 4 Mbit/s.<sup>17</sup> In 2013, NCOM established a goal of access to high-speed broadband at 100 Mbit/s throughout the country.<sup>18</sup> In 2016, the Ministry of Transport reiterated this goal, further specifying that 100 Mbit/s broadband should be available to at least 90 percent of Norwegian households by 2020 and to all households in the long-term.<sup>19</sup>

#### Russia

In 2012, the Ministry of Communications and Mass Media of the Russian Federation announced national broadband targets.<sup>20</sup> The Ministry plans to make 100 Mbit/s ultra-fast broadband (UFB) available to 80 percent of Russian residents by 2018, with the ultimate goal of providing affordable and accessible broadband to 95 percent of households by 2020.<sup>21</sup>

#### Sweden

In 2004, a working group within the Kingdom of Sweden's Ministry of Enterprise and Innovation defined broadband both as "IT infrastructure with high capacity" and as "an 'always on' Internet connection with relatively high capacity."<sup>22</sup> The working group noted, however, that "there are market solutions in which a broadband connection does not necessarily automatically mean a connection to the Internet."<sup>23</sup>

In 2011, the Ministry of Enterprise, Energy and Communications established a broadband speed goal of 100 Mbit/s by 2020 for 90 percent of permanent Swedish households and businesses.<sup>24</sup> Sweden believes that Internet access "makes it possible for people to fully exercise their rights to freedom of expression and opinion," and thus "States have a responsibility to ensure that Internet access is broadly available."<sup>25</sup>

#### **United States of America**

The United States of America's Federal Communications Commission (FCC) defines broadband as "highspeed Internet access [that] allows users to access the Internet and Internet-related services at significantly higher speeds than those available through 'dial-up' services."<sup>26</sup> Although the FCC does not generally define broadband in terms of data throughput,<sup>27</sup> it has adopted various broadband speed thresholds for specific regulatory purposes. For example, the FCC is required to evaluate annually whether "advanced telecommunications capability"<sup>28</sup> is being deployed to all Americans in a "reasonable and timely" fashion.<sup>29</sup> In 2016, the FCC conducted this analysis using actual download speeds of 25 Mbit/s and actual upload speeds of 3 Mbit/s.<sup>30</sup>

In March 2010, the FCC released a comprehensive plan for broadband deployment in the United States.<sup>31</sup> That plan, referred to as "The National Broadband Plan," recommends that the government establish a goal of at least 100 million U.S. homes with affordable access to actual download speeds of at least 100 Mbit/s and actual upload speeds of at least 50 Mbit/s "over the next decade."<sup>32</sup> In Alaska,<sup>33</sup> the state's Statewide Broadband Task Force<sup>34</sup> established a similar goal of 100 Mbit/s connectivity (up and down) for every Alaska household by 2020.<sup>35</sup>







Broadband is a transformative platform that affects nearly every sector of society, including, but not limited to, the economic, educational, governmental, health, scientific and public safety sectors. In addition to the macro-level benefits discussed below, broadband also enhances individual's lives on a micro-level, allowing them to interact and prosper (whether economically, educationally, politically or culturally) in ways not possible before the advent of broadband.

#### 1. Economy

In terms of economic benefits, a 2009 study by the World Bank found that for every 10 percent increase in broadband penetration, there is a corresponding 1.21 percent increase in per capita GDP in high-income economies.<sup>36</sup> The rate of per capita GDP increase is even more significant in developing countries- 1.38 percent.<sup>37</sup> Other studies by the World Bank show that the deployment of broadband facilitates the adoption of more efficient business processes, cuts transaction costs, and provides better access to labor pools, materials and consumers.<sup>38</sup> Companies adopting broadband-based processes improve labor productivity by five percent in the manufacturing sector and by 10 percent in the services sector.<sup>39</sup> Studies also suggest that 10 percent higher broadband penetration in a specific year correlates to 1.5 percent greater labor productivity over the following five years.<sup>40</sup> In addition, the deployment of broadband accelerates innovation by introducing new technologies, services and business models.<sup>41</sup> It enhances human capital by facilitating knowledge and technical skill acquisition."42

#### 2. E-Commerce

Broadband allows for new business models and ways of doing business. Perhaps one of the most widespread and important applications of broadband Internet is e-commerce, which includes business-to-business transactions (e.g. digital advertising, invoices, recruitment, financing) and customer-to-business exchanges (e.g. "e-tailing," bill payments, electronic reservations and ticket purchases, and web-based banking).<sup>43</sup> Indeed, broadband Internet access may offer the best mechanism for advertising Arctic goods and services to the rest of the world. Substantial increases in demand are possible for Arctic businesses able to take advantage of broadband to advertise their goods and services to a broader base of consumers. Broadband also offers a means to accept payments from far-flung individuals as well as purchase supplies and equipment for Arctic businesses themselves.

### 3. Education

Broadband has the potential to transform education by giving teachers and students access to learning resources and technologies that will allow them to gain the knowledge and skills necessary to live and work in the 21st Century.<sup>44</sup> For example, broadband facilitates distance learning by connecting teachers to students and allows greater access to a wide range of educational materials not available locally. With reliable broadband access, educators can create diverse virtual learning environments as well as more personalized and interactive curricula. Broadband enables students to communicate with, and learn from, their counterparts and instructors around the world.

#### 4. E-Government

Broadband enables e-government, which refers to the use of information and communication technology to enhance the range and quality of public services to citizens and businesses while making government more efficient, accountable and transparent.<sup>45</sup> E-government expands access to government services and information. For example, e-government may allow citizens to find information about identification documents and allows them to apply for those documents online. In addition, e-government redefines the relationship between citizens, businesses and governments by creating innovative platforms to prioritize and craft policy, reform public services and reduce regulatory burdens.<sup>46</sup>

#### 5. Government Participation and Community Involvement

Broadband helps to prevent political isolation and to enable political engagement and participation. Broadband improves access to information and allows individuals to take part in the political process regardless of one's physical location. Broadband also improves community involvement by allowing stakeholders to exchange ideas, meet, and collaborate in order to work through issues confronting the community. Moreover, broadband facilitates programs designed to foster increased public, social, and economic participation of individuals with physical and cognitive disabilities, unemployed youth, the economically inactive, immigrants and the elderly.<sup>47</sup>

#### 6. Medicine/Public Health

Broadband enables the development of telemedicine, which can have a transformative effect on otherwise remote regions. The expansion of robust broadband services creates the foundational infrastructure that allows health care providers to improve the health of their patients regardless of where they live. Telemedicine allows for long-distance consults and diagnoses, real time medical monitoring, and secure electronic health information exchange. In addition, given the shortage of some medical specialties in rural and remote parts of the Arctic, telemedicine can play an important role in ensuring patients in rural communities get access to the specialty care they need.48

#### 7. National Interests

Access to secure broadband is critical to national interests such as state security and defense. Climate change combined with estimates suggesting that the Arctic seabed may hold substantial oil and natural gas reserves may lead to increased economic and military operations in the Arctic.<sup>49</sup> Given the likelihood of increased communications support requirements arising from such operations, expanded broadband deployment should be a priority.





More generally, the ability to easily locate and quickly exchange data, including videos and other large files, is essential to ensuring the security of the nation-state as well as the safety of citizens at home and abroad. Allowing key stakeholders to rapidly engage and respond to developing situations or threats quickly is a significant benefit of broadband. Broadband also provides a window into the activities of anti-government groups or individuals and supports numerous technologies critical to national defense.

#### 8. Scientific Research

Broadband enables new and more robust ways of conducting scientific research. Researchers rely on connectivity to access information, share scientific resources and instruments, and virtually collaborate with colleagues around the world.<sup>50</sup> For example, scientists conducting field research in the Arctic require broadband connectivity to access remote computing facilities and transmit field data. Particularly as scientific endeavors increasingly rely on data, simulation and models, e-infrastructures enable new ways of scientific collaboration and resource sharing, as well as wider availability of repositories of scientific data.<sup>51</sup> Broadband may also facilitate knowledge exchange between indigenous and local populations, industry and researchers.

#### 9. Trade and Transportation

Broadband is essential to trade and transportation in the Arctic. Approximately 80 percent of the world's industrial production takes place north of the 30th parallel north, making the Arctic Ocean a shortcut between the world's most advanced and productive regions.<sup>52</sup> Compared to traditional routes between the North Pacific and North Atlantic Oceans, advantages of trade and transportation routes through the Arctic Ocean include shorter distances, reduced fuel consumption and carbon emissions, faster deliveries of goods and higher profits.<sup>53</sup> As the Arctic becomes increasingly important to the global economy, access to real-time information regarding weather and ice conditions is critical for navigation safety, efficiency of travel and regulatory compliance.

Broadband is already helping to shape the future of the maritime and aeronautical industries. Broadbanddriven connectivity in the maritime industry is improving telematics, telemetry, efficiency, safety, and crew welfare and training.<sup>54</sup> Broadband technology is beginning to expand to leisure ships as well, enhancing the passenger experience. In the aeronautical industry, broadband-based software and communications systems are making it easier for pilots, flight engineers and air traffic controllers to monitor flight operations, safety and security.<sup>55</sup> Broadband is also emerging in the passenger airline industry as a high value customer service that allows passengers to access live television and high-speed Internet while in flight.

#### **10. Public Safety**

Broadband is critical to the public safety community in any region.<sup>56</sup> Access to high-speed Internet facilitates communication and allows for more precise disaster planning and response. It also reduces the cost of operations and allows public safety officials to allocate scarce resources more effectively. In addition, reliable broadband can assist in organizing and analyzing data in a meaningful way so that officials can make realtime decisions in "life and death" situations. Similarly, broadband plays an important role is search and rescue ("SAR") operations. Depending upon the particular circumstances, SAR operations involve a number of communications technologies, including voice radios to guide first responders, satellite phones to expand the reach of voice communications, and GPS signals to help navigate.

### C. UNIQUE CHALLENGES TO EXPANDING BROADBAND IN THE ARCTIC

The Arctic Council (AC) is the leading inter-governmental organization promoting pan-Arctic cooperation. Each of the AC's<sup>57</sup> eight member states has developed a national broadband plan tailored to its specific needs and goals.<sup>58</sup> However, many of the challenges posed by the Arctic's unique characteristics are common among all AC member states.<sup>59</sup> By focusing on these shared challenges, a regional broadband strategy can be developed to complement efforts already underway in individual AC member states.

The following are some of the challenges posed by the Arctic:

- depending on the season, waterways or temporary ice roads.<sup>61</sup>
- affordable communication services to end users challenging.<sup>62</sup>
- maintenance and evolution of broadband technologies more difficult.
- recent technological advances is lacking.66

• <u>Dispersed population</u>: Much of the Arctic is characterized by small, rural settlements separated from one another by large distances. For example, approximately 120 Alaskan villages have fewer than 1,000 residents, and many have fewer than 100 residents.<sup>60</sup> Communities such as these are highly dispersed and are often not connected to any road system, limiting ingress and egress to air and,

Geography: The geographic breadth and challenging terrain of the Arctic make much of the region difficult to service from an economic perspective and also make building, maintaining and providing

Harsh climate: Arctic temperatures are some of the coldest in the world. In Svalbard, Norway, for example, temperatures can drop as low as -46°C in the winter.<sup>63</sup> Among other challenges, these temperatures contribute to a short construction season for deploying the physical infrastructure necessary for other broadband technologies. Large amounts of snow and ice also make the

Higher costs: Energy in some parts of the Arctic is expensive,<sup>64</sup> thus increasing costs associated with building the physical infrastructure necessary for broadband deployment. In addition, the components of the broadband infrastructure may themselves be more expensive. For example, fiber optic cables must be heavily armored in order to withstand the freeze and thaw cycles of the harsh Arctic climate.

Human resource gap: The Arctic has an insufficient network of specialized contractors to install equipment necessary for broadband deployment. Similarly, the number of IT professionals gualified to maintain technology and deal with equipment failures in the Arctic is small.<sup>65</sup> Even in locations where an adequate number of professionals may exist, the training necessary to keep pace with



- Inadequate satellite coverage: Current communications satellites operating in geostationary Earth orbit do not fully cover the Arctic region.<sup>67</sup> Satellite-based broadband in the Arctic can also experience latency issues.68
- Lack of a comprehensive strategy: There is currently no comprehensive strategy for connecting all Arctic communities to one another and the rest of the world.<sup>69</sup> AC member states tend to work in isolation from one another when attempting to address their internal user needs.<sup>70</sup>
- Lack of transportation infrastructure: Due, in part, to the harsh climate and dispersed population of the Arctic, road and railway infrastructure development has been limited. Many Arctic communities receive supplies primarily via sea or air.<sup>71</sup> However, due to the lack of natural harbors in many parts of the Arctic and the short summer season, even sea routes can be limited. Without adequate roads and railways, transporting the heavy equipment necessary to lay terrestrial cable or build towers or other facilities necessary for broadband deployment is challenging.
- Lower adoption rates/demand: Bandwidth tends to be limited and expensive throughout the Arctic. For example, individuals living in Nunavut, Canada pay 45 percent more for broadband than their counterparts in the lower Canadian provinces.<sup>72</sup> The broadband available in Nunavut is also considerably slower than that available in other parts of Canada.<sup>73</sup> The high cost for slow speeds contributes to lower broadband adoption rates in the Arctic. Also contributing to lower adoption rates is the lack of awareness of the Internet in some isolated Arctic communities or the belief that the Internet does not facilitate access to relevant (that is, local or localized) content and services.74
- <u>Undeveloped or underdeveloped investment models</u>: The existing investment infrastructure in the Arctic is not meeting the rapid pace of change and convergence of communication services available in other non-Arctic regions.75

As these challenges demonstrate, a broadband strategy for the Arctic requires a multi-pronged approach focused on ways to promote broadband deployment while encouraging demand to ensure that the benefits of connectivity are fully realized.

# III. The Current State of Broadband in the Arctic

Realizing the ultimate goal of universal broadband in the Arctic requires a dual focus on deploying infrastructure and ensuring adoption across all segments of society, from consumers and businesses to government and civil society institutions. A necessary initial step in the process is assessing the current state of broadband coverage and adoption throughout the various Arctic communities.

#### A. BROADBAND USAGE BY COUNTRY

#### Canada

According to the Government of Canada, over 99 percent of Canadian households currently have access to broadband with speeds of at least 1.5 Mbit/s.<sup>76</sup> However, only 27 percent of households in the rural Nunavut region have access to broadband.<sup>77</sup> The Connecting Canadians program, part of Digital Canada 150, will invest up to \$305 million to extend broadband access throughout the country with the goal of bringing speeds of at least 5 Mbit/s to an additional 280,000 homes in rural and northern regions of the country.<sup>78</sup>

#### Denmark

In 2014, the European Commission estimated that overall broadband coverage in the Kingdom of Denmark stood at 99.5 percent of Danish households.<sup>79</sup> Between 2012 and 2014, Denmark increased its rural DSL coverage by six percent, rural VDSL coverage by over 20 percent, rural LTE coverage by 96.8 percent, and overall rural broadband coverage by 1.3 percent.<sup>80</sup> Networks capable of providing at least 30 Mbit/s were available to 91.7 percent of homes overall and 53.7 percent of rural households.<sup>81</sup>

Regular Internet users account for 93 percent of Denmark's population; 75 percent of those users have basic digital skills that enable them to interact online and consume digital goods and services.<sup>82</sup>







According to the European Commission, some form of broadband was available to 99.8 percent of Finnish homes as of 2014.83 Between 2012 and 2014, Finland increased its rural DSL coverage by 7.9 percent and rural LTE coverage by 54 percent.<sup>84</sup> Overall rural broadband coverage stayed the same between 2012 and 2014. Networks capable of providing at least 30 Mbit/s were available to 75.1 percent of homes overall and 7.5 percent of rural households.<sup>85</sup>

Ninety-one percent of the Finnish population uses the Internet regularly, and 75 percent have basic digital skills that enable them to interact online and consume digital goods and services.<sup>86</sup>

#### Iceland

In 2014, the European Commission estimated that overall broadband coverage stood at 98.7 percent of Icelandic homes.<sup>87</sup> Between 2012 and 2014, Iceland increased its rural DSL coverage by over seven percent, rural VDSL coverage by 69.1 percent, rural LTE coverage by 46.9 percent, and overall rural broadband coverage by 2.6 percent.<sup>88</sup> Networks capable of providing at least 30 Mbit/s were available to 89.6 percent of homes and 70.2 percent of rural households.<sup>89</sup>

Overall, 97 percent of Icelandic people use the Internet and 83 percent of them have basic digital skills that enable them to interact online and consume digital goods and services.<sup>90</sup>

#### Norway

As of 2014, 99.3 percent of Norwegian homes had access to some form of broadband coverage according to the European Commission.<sup>91</sup> Between 2012 and 2014, Norway increased its rural VDSL coverage by 7.9 percent, rural LTE coverage by 28 percent, and overall rural broadband coverage by 6.1 percent.<sup>92</sup> Networks capable of providing at least 30 Mbit/s were available to 78 percent of homes overall and 31.2 percent of rural households.<sup>93</sup> In the Arctic regions of Troms and Finnmark, broadband is available via fiber networks as well as mobile wireless. Wireless providers TeliaSonera and Telenor are expanding 4G coverage in the region and full 4G coverage is expected by end of 2017.94 In addition, a study is underway to map the opportunities for broadband in Norway's maritime Arctic territories.95

Overall, 96 percent of the Norwegians use the Internet and 80 percent of them have basic digital skills that enable them to interact online and consume digital goods and services.<sup>96</sup>

#### Russia

A 2015 study by the World Bank indicates that fixed broadband service is available to 56.5 percent of all Russian households.<sup>97</sup> In the second quarter of 2013, 21 percent of broadband connections were at speeds above 10 Mbit/s.<sup>98</sup> More than 56 percent of Russian households overall, and 33.3 percent of rural Russian households, had broadband access at speeds of 256 Kbit/s or higher in October 2013.99 As of the third guarter of 2015, about 87 percent of Russian Internet users enjoyed connection speeds above 4 Mbit/s.<sup>100</sup>

#### Sweden

In 2014, the European Commission calculated that 99.5 percent of Swedish households had access to some form of broadband.<sup>101</sup> Between 2012 and 2014, Sweden increased its rural DSL coverage by 0.6 percent, VDSL coverage by 0.3 percent and rural LTE coverage by 27.4 percent.<sup>102</sup> Networks capable of providing at least 30 Mbit/s were available to 76.4 percent of homes overall and 13.9 percent of rural households.<sup>103</sup>

Overall, 89 percent of Swedish people use the Internet and 72 percent have the basic digital skills that enable them to interact online and consume digital goods and services.<sup>104</sup>

#### United States of America

As of 2016, the Federal Communications Commission estimated that 90 percent of Americans have access to broadband with 25 Mbit/s of upload speed and 3 Mbit/s of download speed.<sup>105</sup> Sixty one percent of Americans living in rural areas have access to 25 Mbps/3 Mbps broadband.<sup>106</sup>

Currently, 126 Alaskan communities have broadband access via technologies ranging from fiber to fixed wireless service.<sup>107</sup> According to the State of Alaska Department of Commerce, 70 percent of rural Alaskans and 87 percent of non-rural Alaskans subscribe to Internet service.<sup>108</sup> As of October 2014, 91 percent of Alaskan households have broadband access at 3 Mbit/s download and 768 Kbit/s upload.<sup>109</sup> Since 2011, the availability of broadband service at speeds of 50 Mbit/s download and 1.5 Mbit/s upload has increased by 61 percent and service at 100 Mbit/s download/1.5 Mbit/s upload has increased by approximately 60 percent.<sup>110</sup>







### **B. AVAILABLE TECHNOLOGIES**

Deploying broadband in the Arctic does not necessarily require the invention of new technologies. Indeed, numerous technologies already exist that could facilitate Arctic broadband deployment. These technologies, along with some of their technical advantages and disadvantages, include:

#### **Optical fiber**

- Pros: Very fast and efficient with less signal degradation; highly resistant to electromagnetic interference and low rate of bit error; scalable (wavelengths can be turned on or off on demand); longevity once deployed and relatively easy to replace or repair.
- **Cons**: High investment costs; time consuming, lengthy deployment; susceptible to physical damage, such as accidental cutting; unidirectional light propagation requires concurrent cables to be laid to achieve bidirectional propagation of information.
- Mobile Wireless (3G, 4G, 4G/LTE, 5G)
  - Pros: More easily deployed than fix broadband; provides the convenience of mobile broadband access.
  - Cons: Signals are shared and thus can be degraded at peak usage times; signal strength impacted by distance, weather, line-of-sight, and other external factors. In addition, with regard to 5G specifically, the technology is not yet ready for deployment. The ITU aims to finalize 5G standards and protocols by 2020.<sup>111</sup> 5G also requires high-speed terrestrial backhaul.
- **Microwave** (Fixed Wireless)
  - Pros: Wireless backhaul alternative to copper or fiber connections (less expensive, easier deployment); lower installation costs compared to buried cable systems; can transfer data at speeds of 1 Gbit/s.
  - **Cons**: Limited range compared to satellite; requires line of sight; may be subject to rain fade; bandwidth is shared.

#### Satellite

Pros: Terrestrial backbone and area networks are not needed; ability to connect users across large regions; heavily utilized by the maritime industry; LEO and other satellites with non-geostationary orbits that are currently being developed and deployed could provide the Arctic with broadband speeds of 50 Mbit/s.

- **Copper** (e.g., Digital Subscriber Line (DSL))
  - less so in the Arctic region).
  - DSL speeds).
- **Coaxial cable** (via cable TV networks)
- generally more capable/efficient than traditional copper telephone networks.
- peak hours; often not available in remote areas.





**Cons**: Many existing satellites are geostationary (positioned over the equator) and service degrades substantially at latitudes above 75 degrees north; expensive initial investment and maintenance; expensive end-user devices; signal degraded by weather and line-of-sight problems.

**Pros**: Rely on existing copper wires that are already present in many locations (though potentially

**Cons**: Speed depends on the length of the copper line; upload speeds generally substantially lower than download speeds; slower than fiber (though VDSL and other developments increasing

Pros: Rely on existing cable TV networks (lower deployment costs where networks already exist);

Cons: Internet use by multiple users simultaneously can degrade service and slow speeds during



# IV. Funding Options

A number of obstacles exist with respect to obtaining adequate funding for infrastructure development in the Arctic. These include the region's dispersed population, expansive geography, extreme weather conditions and difficult terrain, all of which undermine business cases focused on high investment returns. As a result, an effective strategy should focus on a variety of funding options.

#### A. PRIVATE INVESTMENT

Concerns over return on investment-due to the high cost of deployment in challenging and expansive terrain combined with a small and sparsely populated customer base-often deter private investment in broadband deployment in the Arctic. However, increasing reliance on Arctic shipping routes, expanded oil and gas exploration, and other business activities in the region-as well as the addition of new anchor institutionsmay provide increased incentives for improved access to private finance.

In addition to profitability and return on investment, private investment depends heavily on the regulatory climate.<sup>112</sup> The Organization for Economic Co-operation and Development identified particular policy initiatives that may promote private investment in broadband, including policies to undertake the following:

- Improve access to passive infrastructure (conduit, poles, and ducts) and coordinate civil works as an effective means to encourage investment.
- Encourage and promote the installation of open access to passive infrastructure when public works are undertaken.
- Ensure access to rights-of-way in a fair and nondiscriminatory manner.
- Allow municipalities and/or utilities to enter telecommunications markets. Where market distortion is a concern, policy makers could limit municipal participation to basic investments (such as the provision of dark fiber networks under open-access rules).
- Provide greater access to spectrum and adopt more market mechanisms to promote more efficient spectrum use.<sup>113</sup>

#### **B. PUBLIC-PRIVATE PARTNERSHIPS**

Recognizing that both the public and private sectors can benefit from pooling their financial resources, knowhow and expertise, public-private partnerships (PPPs) can be an effective tool to promote delivery of services (including broadband) to all citizens.

PPPs are a widely used and flexible funding mechanism, embracing a range of different cost and risk-sharing arrangements.<sup>114</sup> For example, PPPs for broadband development and deployment may be private-sector lead, whereby a commercial operator builds, owns and operates the network; government-sector lead, whereby a public entity or municipality owns the network and private partners construct, operate and/or maintain the network in exchange for financial and in-kind support; or follow a joint-ownership model whereby a commercial operator and the public enterprise jointly invest in the network and share capacity.<sup>115</sup>

PPPs can also take the form of a regulated monopoly. Mexico, for example, has authorized a wholesale "carrier's carrier" network for the deployment of a new fixed and wireless national broadband network. The entity selected to build out the network will be awarded a 20-year public-private partnership to build an LTE network that must cover a minimum of 85 percent of the population.<sup>116</sup> Once complete, the entity awarded the exclusive right to build out the network will then have the exclusive right to sell capacity at fixed rates to any requesting service provider.<sup>117</sup> The network could potentially be used by hundreds of mobile-service providers, but its operator would be barred from providing service directly to consumers.<sup>118</sup>

Regardless of the form of the PPP, governments and regulators committed to broadband deployment can take several concrete steps to encourage private investment. These include:

- could include specific rules for setting service rates.
- as mechanisms under which they can be changed.<sup>119</sup>
- may unduly impede broadband deployment.
- promote private investment.

Several key issues should be considered before initiating PPPs to expand broadband deployment, however.

First, after careful examination, governments should determine whether private sector investment alone is unlikely without government support.

Second, governments should recognize that deployment does not equal adoption. Providing individuals and businesses with broadband access does not mean that they will subscribe. Thus, to help ensure the success of the PPP, programs encouraging broadband adoption should accompany broadband deployment.



• "Hard-wiring" various aspects of broadband policy and regulation in legislation. For example, the law

Delegating implementation of more general legal provisions to an independent regulatory body.

• Using legally binding license conditions that set out the rights and responsibilities of providers as well

Identify and address regulatory barriers, such as permitting requirements and rights-of-way rules that

Create tax incentives, such as tax credits, tax certificates and accelerated depreciation schedules to

Third, governments should understand that private companies expect a reasonable return on their investment in broadband infrastructure and that private companies will assess the regulatory and other costs placed by governments on broadband deployment projects. Projects expected to have low rates of return will have trouble attracting private capital on favorable financial terms.

Fourth, careful consideration should be given to developing technology-neutral PPPs. Many PPPs fail because they assume that one technology is the only solution.<sup>120</sup> For example, assuming that fiber is the only solution ignores the fact that remote areas may be more effectively served by microwave technologies that have a lower CAPEX per subscriber than wired technologies.<sup>121</sup> PPPs for broadband deployment should incorporate, or at the very least consider and evaluate, a range of technologies suitable to the goal at hand.<sup>122</sup>

#### C. GOVERNMENT INVESTMENT

Relying upon government support alone to fund Arctic broadband deployment may present a challenge. Because each of the AC member states has developed a national broadband plan (in one form or another), government funding contributions are likely to be largely dictated by the strategies, preferences and politics of the various member states.<sup>123</sup>

Nevertheless, government support of Arctic broadband deployment could take a number of forms, including: subsidies for use in deploying infrastructure (loan guarantees, grants, etc.); subsidies for consumers to promote demand; tax incentives to encourage broadband deployment; and creation/expansion of universal service funds. Government support for Arctic broadband deployment could also come from sovereign wealth funds, which are investment funds owned by the governments of sovereign states and funded mainly by foreign exchange and reserve assets.<sup>124</sup>





## V. Past, Current, and Proposed Projects

### A. FAR EAST FIBER

In 2014, state-owned Russian telecom corporation Rostelecom began construction of an 1,855 km fiber optic cable line that will link the regions of Kamchatka, Sakhalin and Magadan in Far-Eastern Russia. The project is a part of a Russian federal program designed to bridge the digital divide by providing broadband access points to settlements with populations between 250 and 500 people. The network will have a maximum capacity of 400 Gbit/s in each direction, with the ability to expand up to 8 Tbit/s.

Phase one of the project was completed in 2015 and connects Sakhalin with Magadan via a terrestrial fiber network. Phase two will connect Okha. Sakhalin to Ust-Bosheretzk. Kamchatka via a submarine cable in the Okhotsk Sea. Rostelecom contracted with Huawei Marine to construct the 900 km submarine fiber cable connecting Sakhalin and Kamchatka. Phase 2 is scheduled for completion in the first quarter of 2017.

Additionally, Rostelecom is considering constructing a submarine fiber optic cable connecting Sakhalin with the Kuril Islands of Iturup, Kunashir and Shikotan. The cable would be 940 km long with a bandwidth of 40 Gbit/s. The scheduled in-service date would be in 2018.

#### **B. GREENLAND CONNECT**

Greenland Connect is a 4,500 km submarine cable that connects Canada, Greenland and Iceland. The cable was put into operation in March of 2009 and expanded in the summer of 2010.<sup>125</sup> Greenland Connect contains two fiber pairs specified for 128\*10 Gbit/s each and two additional fiber pairs specified for 10 Gbit/s each.<sup>126</sup>





TELE Greenland A/S ("Tele-Post"), which is Greenland's sole telecommunications provider and which is whollyowned by Greenland's government, carried out the project and owns and operates Greenland Connect.<sup>127</sup> The project was funded via a €20 million grant from the Nordic Investment Bank.<sup>128</sup>

Today, Greenland Connect serves as a vital part of Greenland's telecommunications infrastructure. Tele-Post provides fixed broadband to Greenland's population using a three zone pricing model; people living in more populated areas pay higher prices for service in order to subsidize the cost of providing fixed broadband to more rural areas. Tele-Post offers mobile broadband based on a "solidarity pricing" model - prices are uniform throughout the island so that citizens share the burden of paying for the infrastructure necessary to provide service to remote areas. Over 99 percent of Greenland's population has access to mobile broadband. Greenland's home rule government views the island as a "natural monopoly" and rejected proposals to liberalize the telecommunications market.

#### C. MACKENZIE VALLEY AND DEMPSTER HIGHWAY FIBER

The Mackenzie Valley Fiber Link (MVFL) project is a Government of the Northwest Territories, Canada (GNWT) initiative to provide a state-of-the-art high-speed fiber optic telecommunications link along the Mackenzie Valley and Beaufort Delta.<sup>129</sup> The MVFL project consists of the installation of 1,154 km of high-speed fiber optic telecommunications cable from McGill Lake to Inuvik, connecting communities along the Mackenzie Valley with Canada's high-speed fibre optic backbone network.<sup>130</sup> The fiber route will be extended from Inuvik to Tuktoyaktuk following the completion of the Inuvik-Tuktoyaktuk Highway.<sup>131</sup>

In October 2014, the GNWT entered into a 20-year Project Agreement with Northern Lights General Partnership (NLGP) to deliver the project.<sup>132</sup> NLGP is a consortium comprised of Ledcor as the construction partner and NorthwesTel as the service provider.<sup>133</sup> The project is expected to cost approximately \$84 million.<sup>134</sup>

Relatedly, the Yukon government is planning a fiber optic line along the Dempster Highway to Inuvik, Northwest Territories.<sup>135</sup> The \$32 million cable will connect to the Mackenzie Valley Fiber Link and will end frequent Internet service disruptions in the territory by providing it with a backup link to Southern Canada.<sup>136</sup>

#### **D. MIDGARDSORMEN**

Midgardsormen is a Norwegian-lead project seeking to design, build and operate a Norway-centric trans-Atlantic cable system to connect Norway and Sweden to the East Coast of the United States. Specifically, Midgardsormen will connect Virginia Beach, Virginia to Blaabjerget, Denmark, with a possible connection to Mo i Rana, Norway. Currently, the basic network architecture is in place, and backhaul and vendor negotiations are ongoing. Vendor selection was scheduled for June 2016, with a service goal of August 2018.

#### E. NORILSK FIBER OPTIC BROADBAND

Norilsk is one of the biggest cities in the Arctic and the only Russian city above the Arctic Circle with a permanent population of more than 175,000. The city currently lacks fiber optic broadband; Internet communications are handled by satellite with a general throughput of 1 Gbit/s. Nornickel, a privately-owned Russian mining and metallurgical company, has committed to constructing a 957 km long fiber optic cable that will provide Norilsk with broadband at speeds up to 40 Gbit/s. The cable will be laid through the Yamal-Nenets Autonomous Region and Krasnoyarsk Territory. Residents living along the line will be able to connect to it, as will inhabitants of Igarka, Snezhnogorsk and employees of the Vankor oil and gas field. Future infrastructure development could extend the cable to Yamal.

Although the project will not be profitable, Nornickel has agreed to construct the network as a social benefit to its many employees in Norilsk and as a means to develop its own IT infrastructure. Once construction is finished, Nornickel may be able to offset some of its cost by selling or leasing access to the cable to fiber optic operators. The cable is currently scheduled to be complete in 2017.

#### F. NXTVN'S OULU NORDIC EXPRESS

This will be a cross-border, Nordic-centric, Gulf of Bothnia bridge connecting cities in the Nordic regions of Finland and Sweden to Norway with onward connections to mainland Europe via submarine and terrestrial networks. NxtVn's Oulu Nordic Express is estimated to cost approximately \$80 million dollars (depending upon the number of landings). NxtVn may seek funding for the project from the European Commission.

#### G. NUVITIK

Nuvitik Communications, based in Ottawa, Ontario, specializes in submarine fiber optic cable operations. Its mission is to bring a telecommunications infrastructure to Canada's North.<sup>137</sup> The company is working on a proposed submarine cable<sup>138</sup> network that will provide fast, reliable cable Internet for hospitals, schools, government, authorities, businesses and homes in Canada's North.<sup>139</sup> The Ivaluk Network project was announced in January 2014, and the proposed network would provide 10 Gbit/s wavelengths per fiber pair with a design capacity of up to 64 waves per fiber pair.<sup>140</sup>

The Ivaluk Network project involves three sections-Nunavik, Southeast Nunavut, and Northwest Nunavutwith each section connecting every coastal community on its path.<sup>141</sup> Installation includes two phases, with the Nunavik and Southeast Nunavut network routes operational by fall of 2016 and the Northwest installation planned for 2017.142





Quintillion is an Anchorage, Alaska-based company that will build and operate a privately-funded subsea fiber optic system that will initially run 1,850 km along the north and west coast of Alaska.<sup>143</sup> The project has three phases: (1) the Alaska segment; (2) the Pacific segment; and (3) a segment connecting Alaska to Europe via the Canadian Arctic. The Alaska segment, which includes the installation of a 1,850 km subsea fiber optic cable from Prudhoe Bay to Nome, is underway and expected to be completed in early to mid-2017.<sup>144</sup> The Pacific segment will connect Alaska to Japan, with options for additional spurs in Alaska.<sup>145</sup> Phase three will extend the subsea system from Alaska through the Northwest Passage to the United Kingdom, with spurs into select communities in the Canadian Arctic.<sup>146</sup>

Once all three phases are complete, the cable route will enable Internet data to flow between Europe and Asia at speeds as fast as 30 terabits per second.<sup>147</sup>

#### I. RUSSIAN OPTICAL TRANS-ARCTIC SUBMARINE CABLE SYSTEM (R.O.T.A.C.S.)

The Russian Optical Trans-Arctic Submarine Cable System ("R.O.T.A.C.S.") is a Russian-led project that began in the year 2000 but that has been delayed for economic reasons. R.O.T.A.C.S. was developed by the Russian company Polarnet with an estimated cost of \$750 million. The planned 14,700 km long cable would link Japan and London through the Northeast Passage (Northern Sea Route) and provide a capacity of 100 Gbit/s.

#### J. SVALBARD UNDERSEA CABLE SYSTEM

The Svalbard Undersea Cable System provides a broadband connection between Svalbard and mainland Norway. The system consists of twin undersea fiber optic cables—one 1,375 km long, and the other 1,339 km long. In 2010, all houses and apartments in Longyearbyen received a fiber connection, including telephony, Internet, and Internet protocol television ("IPTV"), with Internet access speeds up to 500 Mbit/s. Businesses were connected to fiber-based services in 2013. In addition, 4G wireless broadband is available in Longyearbyen.

The Svalbard undersea cable system was financed and built by Norsk Romsenter Eiendom, the predecessor of Space Norway. Between 2009 and 2014, Telenor invested 12.1 million krone in the project. The system itself is owned by Space Norway AS but operated by Telenor Svalbard.

#### K. TERRESTRIAL FOR EVERY RURAL REGION IN ALASKA ("TERRA")

Terrestrial for Every Rural Region in Alaska ("TERRA") is a hybrid fiber optic-fixed microwave network that has brought broadband services to more than 70 communities in western Alaska for the first time. The network was built by GCI Communication Inc. ("GCI").

In 2010, GCI affiliate United Utilities, Inc. was awarded a \$44.2 million loan and a \$44 million grant from the federal Broadband Initiatives Program<sup>148</sup> and a smaller grant through the State of Alaska. GCI leveraged this public support with its own private investment and since 2010 has been steadily expanding and upgrading the network. In 2011, the TERRA network served 9,089 households and 748 businesses. In 2012, GCI completed the first phase of TERRA to connect southwest Alaska to the fiber backbone in Anchorage. In November 2014, GCI extended the service to Nome and Kotzebue, north of the Arctic Circle. The next phase, which will "ring" the existing network to increase resiliency and capacity, is currently underway.







## VI. Goals And Recommendations

A regional broadband strategy encompassing eight countries with varying needs and degrees of development requires ambitious but flexible objectives. The following recommendations seek to establish long-term, comprehensive, and aspirational goals for the Arctic region, while also allowing for flexible shortterm benchmarks for individual member states in order to strike an effective balance between pragmatic development considerations and the desire to advance a comprehensive strategy for the Arctic region as a whole. At the same time, technology continues to evolve and progress. As such, the process of developing and tailoring broadband strategies for the Arctic should be on-going and iterative. As technology and needs change, so should these goals and recommendations.

The private sector, NGOs, and local, indigenous, provincial and national governments should work together to spur more private sector investment in broadband deployment and facilities. To do so, these entities should conduct comprehensive reviews of laws, regulations, and business practices that could inhibit or frustrate private sector investment in, and deployment of, broadband technologies. Examples of potential legal, regulatory, and business practice reforms and strategies include:



- areas, could also be offered.
- broadband facilities.
- border deployment and data flow.
- environment and archaeological, historic and cultural sites.
- other assets where practicable.
- rollout of new wireless services in the region.

When and where a realistic business case cannot be made for purely private risk capital ventures,<sup>150</sup> governments should work in collaboration with the private sector to identify and implement policies to support efficient and mutually beneficial public-private partnerships designed to support broadband deployment and adoption efforts. Solutions could involve subsidies, loans, grants, and other mechanisms to foster the construction of new broadband facilities and broadband adoption educational efforts.

Furthermore, if public-private partnerships are not practicable, governments should use public resources to finance and operate broadband facilities to offer services on a carrier-neutral and cost-based wholesale basis to help spur adoption. As part of such efforts, governments and private entities should work collaboratively to examine whether a transition path to privately owned and operated network facilities and services can be found.



Modernize tax structures to incentivize private entities to put forth risk capital in the remote and insular areas of the Arctic. Specifically, governments could offer tax incentives tailored to stimulate Arctic broadband investment such as: accelerated depreciation schedules for enumerated technologies; allowing some capital expenditures to be treated as expenses; exempting specific technologies and expenditures from sales and use taxes; tax credits; and others.<sup>149</sup> Other incentives, such as grant programs or reduced regulation in exchange for a commitment to broadband deployment in remote

Streamline local zoning and permitting laws to allow for faster and easier construction of new

Collaborate with international partners (government, private sector, and NGOs) to facilitate cross-

Partner with environmental protection and land management agencies and NGOs to work together on best practices to allow for the most rapid broadband deployment possible while protecting the

Coordinate with government-owned and influenced assets and anchor institutions such as military installations, research centers, schools, libraries, health care clinics and community centers to leverage their presence to spur additional broadband deployment to outlying communities. Furthermore, make more readily available government controlled assets, including the following: rights-of-way, subsea cable landing rights on government lands, poles, conduits, roof rights and

Make available more usable and appropriate spectrum in Arctic regions, whether on a licensed, unlicensed and/or shared basis. Additionally, streamline spectrum allocation processes to speed the

Supporting the rapid deployment of a variety of broadband technologies is the crucial first step in getting the Arctic connected to the rest of the world. Deployment without adequate adoption rates will be meaningless, however. Accordingly, the private sector, NGOs and governments alike should work collaboratively to create innovative programs to spur rapid adoption of broadband technologies. Among many ideas to be considered are:

- Address barriers to adoption, such as costs. For instance, broadband Internet access providers should be encouraged to offer, publicize and actively market several tiers of service at various price points that are affordable to the populations being served. Governments should explore new methods of subsidizing broadband access on a means-tested basis for residential subscribers.
- Tackle digital literacy barriers to broadband adoption. Private sector players, NGOs and governments at all levels should be encouraged to work together to promote digital literacy programs, especially in communities that will gain broadband access for the first time.
- Coordinate with schools, libraries, community centers and other anchor institutions to leverage their presence to spur digital literacy and broadband adoption.
- Form special task forces of governments, NGOs, private enterprises, and indigenous governments and councils to work together to address deployment, adoption, and cultural issues relative to the presence of broadband technologies vis-à-vis the unique attributes of indigenous peoples.

## VII. Conclusion

The Arctic is unique. It offers dramatic beauty, abundant natural resources and a robust indigenous population with valued traditions. The Arctic will also undergo more changes in the 21st century due to increased human activity than perhaps any other region on Earth. The deployment and adoption of a variety of broadband technologies can enable the expanding Arctic population to grow and prosper economically and socially while also protecting the environment and indigenous traditions.

Broadband technologies will continue to advance and change rapidly. These Recommendations are intended to serve as a thought provoking guide to help spur more broadband deployment and adoption in the Arctic region. It is our hope that the Arctic Economic Council will continually work on this endeavor because the goal of making available cutting-edge broadband technologies in the region will always be a work in progress. We encourage others to pick up where this document ends.







# AEC Telecommunications Infrastructure Working Group

The Arctic Economic Council established the Telecommunications Infrastructure Working Group (TIWG) in 2016. The TIWG's mission is to draft, from a business perspective, a transnational broadband strategy for the Arctic region that will help to influence the political, social, and economic decisions made regarding telecommunications and information and communications technology (ICT).

The TIWG consists of telecommunications and ICT expert advisors with diverse backgrounds in industry and Arctic affairs. TIWG advisors hail from each nation state with territory above the Arctic Circle, understand the challenges of broadband deployment in the Arctic and have specific knowledge of broadband technologies.

### Robert McDowell, Cooley LLP / United States of America

McDowell chairs the TIWG and is co-leader of the global communications practice of the international law firm of Cooley, LLP, a partner in the telecommunications, media and technology practice in Washington, DC. Rob is a former commissioner and senior member of the Federal Communications Commission, where he helped steer two of the then-largest wireless auctions in U.S. history and sparked the creation of an international coalition to combat multilateral Internet regulation. He has been an industry and government leader on a multitude of complex issues in the communications field throughout his career. McDowell provides strategic legal, business and public policy advice to clients on important domestic and international matters in the telecommunications, media, technology and digital media industries. He is also a senior fellow at the Hudson Institute, a nonprofit, nonpartisan policy research organization.

George Hungerford, Gwich'in / Canada



Hungerford is senior legal counsel, corporate finance at a Canadian government securities regulator. Previously, he was in private practice at a major international law firm as corporate/commercial counsel for a variety of clients, including First Nations, which he represented in negotiating impact benefit agreements. George was also formerly an investment banker and consultant, focusing on telecommunications and IT, in Silicon Valley, California and East Asia. He is Gwich'in and sits on the Gwich'in Development Corporation Board (Inuvik). Hungerford holds degrees in law from the University of British Columbia, business from Stanford University, and electrical and computer engineering from Queen's University. He also holds a Chartered Financial Analyst charter.

Viet Nguyen, Ericsson Canada Inc. / Canada 🛛 🖗



Nguyen is director, regulatory and government relations for Ericsson Canada Inc. In that role, he is responsible for the company's interaction with government and associations on technical, regulatory and policy issues. Viet was promoted to the director level in 2006, having held the position of manager, director, regulatory affairs and government relations since 2000. Viet began his career at Ericsson in 1988 at the Research & Development and Services Center in Montreal, where he held various positions including system design,

project manager and technical project manager. He relocated to Ericsson's Mississauga office in 1998 assuming the role of manager, business strategy as part of a team responsible for short and long-term corporate strategies in Canada. Currently, Nguyen holds the chair position of the Canadian Electronics and Communications Association (CECA) and is also a member of the Radio Advisory Board of Canada's (RABC) executive committee.

### Oana Spinu, Nunavut Broadband / Canada 🛛 👾

Spinu manages the overall operations of the not-for-profit corporation Nunavut Broadband Development Corporation and its operating subsidiaries. In that role, she performs a variety of functions including strategic planning oversight, policy research and development, management, communications and outreach, funding development, and reporting. Oana regularly liaises with a variety of stakeholders including federal and territorial governments, Inuit organizations, telecommunication service providers, media, civil society and the general public.

### Morten Glamsø, Danish Shipowners Association / Denmark

Glamsø is currently senior advisor to the Danish Shipowners' Association, a trade organization that represents 40 ship owners and two offshore companies. In that capacity, he is responsible for representing and promoting the interests of the Danish shipping industry in the Arctic. Such interests include navigational safety, improvements to the telecommunications infrastructure, and ensuring the physical security of ships and personnel. His work puts him in close contact with member companies, shipping authorities and decisions makers. Before joining the Danish Shipowners' Association, Glamsø was a naval officer in the Royal Danish Navy.

### Peter Bay Kirkegaard, Confederation of Danish Industries / Denmark

Kirkegaard is senior advisor at the Confederation of Danish Industry, which is Denmark's largest business organization. The organization is funded, owned and managed entirely by 10,000 companies within the manufacturing, trade and service industries. He works in the trade policy department, where he focuses on advocacy for multi- and bilateral trade policy. He also works on export controls and Arctic affairs.

### John Strand, Strand Consult / Denmark

Strand's background is in sales and marketing in the IT and publishing sectors. In 1994, he founded Strand Consult, which today focuses on the mobile sector. As founder and principle of Strand Consult, John, along with his team of analysts, evaluates markets and market trends, publishes reports, and holds executive workshops that enable mobile operators and service providers, device manufacturers, and media companies all over the world to maximize investment returns in the mobile sector.







### Suvi Lindén, Pearlcon Consulting / Finland



Lindén is chairperson of the board of NxtVn Finland and vice chair for NxtVn Group. From 2007 to 2011, Suvi was Finland's Minister of Communications responsible for media and telecommunications. She previously served as Minister of Culture, Youth and Sport from 1999 to 2002 and as a member of Parliament from 1995 to 2011. Lindén acted as ITU's special envoy to the Broadband Commission for Digital Development and was also a member of the Broadband Commission from 2010 to 2015. She is a member of the Advisory Board of SocialEco Ltd and an advisor for the Broadband Innovation Fund. In 2011, the Intelligent Community Forum, a New York-based think tank, named her as its "Visionary of the Year" for her commitment to ensuring affordable broadband access to every citizen in Finland.

Heidar Gudjonsson, Vodafone Iceland / Iceland



Gudjonsson is managing director of Ursus and chairman of the board of Vodafone Iceland. Heider founded Eykon Energy, an oil exploration company and is the chairman of the board. He also founded the first infrastructure investment fund in the Arctic. Among other accomplishments, Heidar built hydropower plants in Iceland, organized the largest acquisition of Icelandic fisheries in history, and founded a Norwegian-Icelandic oil exploration company. He is a board member of the Icelandic Arctic Chamber of Commerce, is on the advisory board of the Arctic Circle, and is the author of the "Arctic Push: Iceland and the Opportunities" (2013), a bestseller in Iceland.

Dag Ivar Brekke, Innovation Norway / Norway



Brekke has over 20 years of experience in engineering and infrastructure development and is currently serving as special advisor, Invest in Norway at Innovation Norway. Dag previously served as commercial counsellor at the Royal Norwegian Embassy in Moscow, Russia and as deputy managing director and temporary acting CEO and COO at Store Norske Spitsbergen Grubekompani AS, a government-owned company specializing in Arctic mining and logistics, property development and mineral resources.

Willy Jensen, Retired / Norway



Prior to his retirement in 2012, Jensen served as councillor of electronic communication to the Norwegian Embassy for the European Union. Willy served for 11 years as the director general for the Norwegian Post and Telecommunications Authority, and for two years as the director general to the Norway 2000 Cultural Program. Earlier in his career, he held key positions at the European Space Agency and Spacetec Ltd., and was also a professor at the University of Tromsø. Jensen has served in a number of different capacities on various national and international committees, including as a board member of the University of Tromsø and the Norwegian delegate to the IFIP.

#### Stanislav Dimukhametov, Pro-Arctic / Russia

Dimukhametov is the founder of Pro-Arctic, an organization dedicated to the management, promotion, and responsible and sustainable development of the Russian Arctic. Pro-Arctic also serves as a portal for Arctic professionals to connect to one another and share ideas and resources. In addition, he is responsible for the development of an online geo intelligence platform for the Arctic utilizing GIS applications, geospatial data cloud management, satellite AIS and remote sensing, and scientific data. Stanislav served as director of CreoProm from 2003 to 2012, where he lead the development and implementation of business-to-business, business-to-government, and business-to-community marketing campaigns for leading domestic and international energy, IT, telecom and automotive companies.

#### **Yury Grin**, Intervale Group / Russia

Grin is the deputy director general of Intervale, a group of companies focused on the e-commerce and mobile payment fields. He also serves as vice chairman of the ITU-T focus group on digital financial services. Previously, Yury worked for the ITU as chief of the conference and publication department and deputy to the director & chief of the administration and operations coordination department of the Telecommunication Development Bureau. While at the ITU, Grin assisted developing countries with various telecommunicationsrelated issues including migrating from analog to digital broadcasting, radio frequency management and cybersecurity. Yury has also served in various positions within the Russian Federation's Ministry for Information Technologies and Communications. Yury is the recipient of numerous awards, including the Medal of the Soviet Peace Fund, and is a member of the International Academy of Communications and the Moscow Scientific Society.

#### Marianne Treschow, TreschowConsulting / Sweden

Treschow is the former director general and member of the board of the Swedish Post and Telecom Authority (PTS). She was appointed as the Commonwealth Telecommunications Organisation's ambassador for Scandinavia in 2010-2015. She is also senior advisor to Ericsson Group; advisor-Europe to VoIPSolutions; founder and CEO of TreschowConsulting; senior advisor to EcoDataCenter; expert for the Global Network Women in ICT; and a member of the Royal Academy of Engineering Sciences, the Swedish Institute of International Affairs, and i-CANADA's Governors Council. In addition to a long career in PTS, Marianne has also held a number of important positions in Sweden including director of the Swedish Space Agency, director of the Swedish Natural Sciences Research Council and assistant professor in structural chemistry at Stockholm University.

### Lori Davey, Alaska Communications / United States of America

Davey is director sales and service at Alaska Communications. She previously served as director business development and general manager at Fairweather LLC, which offers medical, aviation, weather forecasting, and logistics services to the oil and gas industries. Lori also serves as a trustee on the Alaska Nature Conservancy.





### Karl Kowalski, University of Alaska Fairbanks / United States of America

Kowalski is chief information technology officer for the Alaska State University system, which consists of 16 campuses spread across 600,000 miles. Karl advises the University president, other university officers and the Board of Regents in matters related to information technology systems and services, and is responsible for the University's core information systems and services infrastructure, as well as for the coordination, development and implementation of system-wide information technology standards.

### Drue Pearce, Crowell and Moring LLC / United States of America

Pearce joined Crowell & Moring in 2010 as a senior policy advisor with their environment, energy and resources, and public policy groups. Drue served in the Alaska State Legislature for 17 years, presiding as Senate President for two terms. She left the Legislature to serve as senior advisor to Secretaries Gale Norton and Dirk Kempthorne at the Department of the Interior. In 2006, she was nominated by President George W. Bush and confirmed by Congress as the first federal coordinator in the Office of the Federal Coordinator for Alaska Natural Gas Transportation Projects, an independent federal agency created by Congress in 2004. Governor Sean Parnell has appointed her to the boards of both the Alaska Gasline Development Corporation and the Alaska Aerospace Corporation. She serves as vice chair of both boards. Pearce also serves on the board of the Marine Exchange Alaska and is the chairman of the board of the Institute of the North.

#### Robert Walsh, General Communications, Inc. / United States of America



Walsh is director of rural broadband development for General Communications, Inc. (GCI). In that role, he is responsible for community relations in rural Alaska, solving sometimes delicate problems and bringing rural perspectives to senior managers. Robert's typical activities include conducting community meetings, following up on complaints and suggestions from the community, supporting GCI's work with the Arctic Council's Task Force on Telecommunications Infrastructure in the Arctic (TFTIA), conducting monthly rural advocacy committee meetings addressing service and transport issues, and creating innovative solutions to improve customers' communications service in the Arctic.

### Kristina Woolston, Quintillion / United States of America



Kristina is vice president external affairs for Quintillion and is responsible for overseeing the company's external affairs. Kristina is also board chair of the Native American Contractors Association (NACA). Kristina was raised in the rural Alaska community of Naknek and is an Athabaskan shareholder of the Bristol Bay Native Corporation, the regional Alaska Native Corporation for southwest Alaska. Kristina worked for the World Trade Center Alaska where she advocated for Alaska businesses in their efforts to break into global markets. She has also worked for a private firm that represented oil and gas developers on the Sakhalin Island projects and for NovaGold Resources on the Donlin Gold project in western Alaska.







### Citations & References

- 1. Alexander Joseph Huber and Josef Franz Huber. UMTS and Mobile Computing 80 (2002).
- 2. In addition to varying definitions, access to broadband is given varying levels of official or legal priority in different nation-states. For example, a "reasonably priced broadband connection" is considered "everyone's basic right" in Finland. Ministry of Transport and Communications, Finland, 1Mbit Internet Access A Universal Service in Finland From the Beginning of July (2010), http://www.lvm.fi/en/-/1-mbitinternet-access-a-universal-service-in-finland-from-the-beginning-ofjuly-782612. In addition, in July 2016, the United Nations Human Rights Council adopted a resolution affirming "the importance of applying a comprehensive human rights-based approach in providing and in expanding access to Internet and requests all States to make efforts to bridge the many forms of digital divides." U.N. GAOR, Human Rights Council, 32nd Sess. at 3, U.N. Doc. A/HRC/32/L.20 (June 27, 2016).
- 3. The definitions and goals discussed in this section provide a "snapshot" in time. The concept of broadband is ever-evolving and iterative.
- 4. FAQs for Canadians, Government of Canada, Digital Canada 150, http://www.feddevontario.gc.ca/eic/site/028.nsf/eng/50010.html.
- 5. Id.
- 6. Internet Speed and Performance, Government of Canada, Canadian Radio-Television and Telecommunications Commission, http://www. crtc.gc.ca/eng/internet/performance.htm.
- Ministry of Business and Growth, Denmark, Better Broadband and Mobile Coverage Throughout Denmark (2013), https://www.evm.dk/ publikationer/2013/13-03-13-bedre-bredbaand-og-mobildaekning-ihele-danmark.
- Ministry of Transport and Communications, Finland, 1Mbit Internet 8. Access A Universal Service in Finland From the Beginning of July http://www.lvm.fi/en/-/1-mbit-internet-access-a-universal-(2010)service-in-finland-from-the-beginning-of-july-782612.
- 9. Country Information Finland, European Commission, Digital Single Market, https://ec.europa.eu/digital-single-market/countryinformation-finland#national-broadband-strategy-and-policy. See also The Ministry of Transport and Communications, Making Broadband Available to Everyone: The National Plan of Action to Improve the Infrastructure of the Information Society (2008), http://www.lvm.fi/ documents/20181/819315/Making+broadband+available+to+everyo ne.+The+national+plan+of+action+to+improve+the+infrastructure+of +the+information+society+(LVM50-2008)/7afa83d5-0f9b-4d78-93da-7731397bc08e?version=1.0.
- 10. Country Information Finland, European Commission, Digital Single Market, https://ec.europa.eu/digital-single-market/countryinformation-finland#national-broadband-strategy-and-policy. A total of EUR 130 million in public aid is available for broadband projects. Of that total, EUR 66 million is state aid, approximately EUR 25 million is from the EU Rural Development Programme for Mainland Finland, and approximately EUR 40 million is from Finnish municipalities. Fast Broadband Project Brings Ultra Fast Internet to Sparsely Populated Areas, Viestintavirasto, Finnish Communications Regulatory Authority, https://www.viestintavirasto.fi/en/steeringandsupervision/ broadband2015.html.

- 11. More Speed for Broadband Universal Service, Ministry of Transport and Communications (Mar. 18, 2015), http://www.lvm.fi/en/-/morespeed-for-broadband-universal-service-796925.
- 12. Ministry of Transport and Communications, Finland, Towards the Internet of Things Broadband Implementation Plan (2016), http:// www.lvm.fi/lvm-site62-mahti-portlet/download?did=206741.
- 13. Id. at 1.
- 14. Press Release, Ministry of Transport and Communications, 1 Mbit Internet access a universal service in Finland from beginning of July (June 29, 2010), http://www.lvm.fi/en/-/1-mbit-internet-access-auniversal-service-in-finland-from-the-beginning-of-july-782612.
- 15. International Telecommunication Union, Workshop on Promoting Broadband, Promoting Broadband: The Case of Iceland, Document PB/08 at 9-10 (2013), https://www.itu.int/osg/spu/ni/ promotebroadband/casestudies/iceland.pdf.
- 16. A survey of Icelandic Internet service providers indicates that broadband speeds between 40 Mbit/s to 500 Mbit/s are available to consumers. See, e.g., Síminn, Individuals, http://vortex.is/internet/ljosleidari/ (last visited June 15, 2016) (noting that Síminn offers Ljósnet - a fiber connection that offers up to 100 MB/s speed connection): Vortex Inc.. https://hringidan.is/internet/ljosleidari/ (last visited June 17, 2016) (noting that Vortex offers optical fiber speeds between 40 MB/s and 500 MB/s); Farice, http://www.farice.is/services/bandwith-capacity/ (last visited June 17, 2016) (noting that Farice offers speeds beginning at 50 MB/s); Vodafone Iceland, https://vodafone.is/internet/internet/ thjonustuleidir-og-verd/ (last visited June 17, 2016) (noting that Vodafone Iceland offers speeds up to 500 MB/s)
- 17. See generally What is broadband?, Norwegian Post and Telecommunications Authority, http://nettfart.no/broadband/about.
- 18. Post-og teletilsynet, Broadband in Norway 8 (2014), http://eng.nkom. no/topical-issues/reports/\_attachment/16031?\_ts=14abf0b9644.
- 19. Ekomplan: Bredbånd og Mobilnett er Like Viktig Som Vei og Jernbane, Samferdselsdepartementet, https://www.regjeringen.no/no/aktuelt/ ekomplan-bredband-og-mobilnett-er-like-viktig-som-vei-og-jernbane/ id2484276/
- 20. The Goals of the Ministry of Telecom and Mass Communications of the Russian Federation 2012-2018, The Ministry of Telecommunications and Mass Media of the Russian Federation, http://2018.minsvyaz.ru/ en/#.
- 21. Carlo Maria Rossotto et al., The World Bank, A Sector Assessment: Broadband in Russia 10 (2015), http://www.worldbank.org/content/ dam/Worldbank/Feature%20Story/ECA/Broadband\_in\_Russia\_Final. <u>pdf</u>.
- 22. IT Policy Strategy Group, Broadband for Growth, Innovation and Competitiveness, The IT Policy Strategy Group: Report by the working group on IT infrastructure and broadband 23 (2006), http://www. government.se/contentassets/8d00e4c30e304eb3b0af5efe3ecf5944/ broadband-for-growth-innovation-and-competitiveness.

#### 23. **Id.**

- 24. Government Offices of Sweden, Ministry of Enterprise, Energy and Communications, ICT for Everyone - A Digital Agenda for Sweden 10 (2011), http://www.government.se/ contentassets/8512aaa8012941deaee5cf9594e50ef4/ict-foreveryone---a-digital-agenda-for-sweden.
- 25. Sweden, Sweden and Human Rights, https://sweden.se/society/ sweden-and-human-rights/.
- 26. Getting Broadband, Federal Communications Commission, https:// www.fcc.gov/consumers/guides/getting-broadband. The FCC has also stated that "Broadband speeds vary significantly depending on the technology and level of service ordered. Broadband services for residential consumers typically provide faster downstream speeds (from the Internet to your computer) than upstream speeds (from your computer to the Internet)." Id.
- 27. Some individual states have defined broadband in terms of speed. For example, in 2014 the Alaska Broadband Task Force defined broadband as "providing two-way data transmission with advertised speeds of at least 768 [Kbit/s] downstream and at least 200 [Kbit/s] upstream to end users, or providing sufficient capacity in a middle mile project to support the provision of broadband service to end users." Statewide Broadband Task Force, A Blueprint for Alaska's Broadband Future App'x B (2014), http://www.alaska.edu/oit/bbtaskforce/docs/Statewide-Broadband-Task-Force-Report-FINAL.pdf.
- 28. "Advanced telecommunications capability" is "defined, without regard to any transmission media or technology, as high-speed, switched, broadband telecommunications capability that enables users to originate and receive high-quality voice, data, graphics, and video telecommunications using any technology." 47 U.S.C. Shgh 1302(d)(1).
- 29. 47 U.S.C. § 1302.
- 30. Inquiry Concerning the Deployment of Advanced Telecommunications Capability to All Americans in a Reasonable and Timely Fashion, & Possible Steps to Accelerate Such Deployment Pursuant to Section 706 of the Telecommunications Act of 1996, as Amended by the Broadband Data Improvement Act, 31 F.C.C. 699 (2016), https://apps. fcc.gov/edocs\_public/attachmatch/FCC-16-6A1.pdf.
- 31. The Federal Communications Commission, Connecting America: The National Broadband Plan (2010). https://transition.fcc.gov/nationalbroadband-plan/national-broadband-plan.pdf.

32. Id. at xiv.

- 33. Alaska is the only U.S. state with land mass above the Arctic Circle.
- 34. The 22-person Task Force was appointed by Commissioner Susan Bell in February 2011 to examine how Alaska should expand highspeed Internet to everyone across Alaska, to review the policies that should encourage broadband development, to determine the cost of needed broadband infrastructure, and to create a blueprint for the future. Statewide Broadband Task Force, A Blueprint for Alaska's Broadband Future 7 (2014), http://www.alaska.edu/oit/bbtaskforce/ docs/Statewide-Broadband-Task-Force-Report-FINAL.pdf.

35 Id. at 8



36. Christine Zhen-Wei Qiang et al., Economic Impacts of Broadband, in 2009. Information and Communications for Development 35 (2009). http://siteresources.worldbank.org/EXTIC4D/Resources/IC4D\_ Broadband\_35\_50.pdf.

37. **Id.** 

- 38. Carlo Maria Rossotto et al., The World Bank, A Sector Assessment: Broadband in Russia 12 (2015), http://www.worldbank.org/content/ dam/Worldbank/Feature%20Story/ECA/Broadband\_in\_Russia\_Final. pdf.
- 39. Id.
- 40. **Id.**
- 41. Id. at 13.
- 42 **Id**.
- 43. Barney Warf, Contours, contrasts, and contradictions of the Arctic internet, 34 Polar Geography 201 (2011). https://www.researchgate. net/publication/241727880\_Contours\_contrasts\_and\_contradictions\_ of\_the\_Arctic\_internet.
- 44. Karl Kowalski, Arctic Telecommunications and Networking -Broadband for a Connected Arctic: Case for a UArctic Thematic Network, ResearchGate, 3 (2015), https://www.researchgate. net/publication/281101160\_Arctic\_Telecommunications\_and Networking\_-\_Broadband\_for\_a\_Connected\_Arctic\_Case\_for\_a\_ UArctic\_Thematic\_Network.
- 45. Nagy K. Hanna et al., National E-Government Institutions: Functions, Models and Trends, in 2009, Information and Communications for development 83 (2009), http://siteresources.worldbank.org.
- 46. See generally Organisation for Economic Co-operation and Development, Recommendation of the Council on Digital Government Strategies, available at http://www.oecd.org/gov/digital-government/ Recommendation-digital-government-strategies.pdf.
- 47. European Commission, Digital Single Market, Digital Inclusion for a Better EU Society, https://ec.europa.eu/digital-single-market/en/ digital-inclusion-better-eu-society.
- 48. Karl Kowalski, Arctic Telecommunications and Networking -Broadband for a Connected Arctic: Case for a UArctic Thematic Network, ResearchGate, 3 (2015), https://www.researchgate. net/publication/281101160\_Arctic\_Telecommunications\_and\_ Networking\_-\_Broadband\_for\_a\_Connected\_Arctic\_Case\_for\_a\_ UArctic\_Thematic\_Network.
- 49. Patrick L. Smith et al., Broadband Satellite Communications for Future U.S. Military and Coast Guard Operations in an Ice-Free Arctic, Aerospace (Jul. 1, 2011), http://www.aerospace.org/crosslinkmag/ summer2011/broadband-satellite-communications-for-future-u-smilitary-and-coast-guard-operations-in-an-ice-free-arctic/
- 50. The UN Broadband Commission for Digital Development, Working Group on Broadband and Science Final Report 5, available at http:// www.broadbandcommission.org/Documents/publications/WG\_ Broadband\_Science\_Final\_Report.pdf.
- 51. Id. at 7.

52. Willy Østreng, The Future of Shipping & Trade in Arctic Waters, World Policy Blog (Apr. 8, 2015), http://www.worldpolicy.org/ blog/2015/04/08/future-shipping-trade-arctic-waters.

#### 53. Id.

54. Mark Holmes, Inmarsat CEO: Aviation Could Become our Biggest Market, Via Satellite Digital (July/August 2016), http://interactive satellitetoday.com/via/july-august-2016/inmarsat-ceo-aviation-couldbecome-our-biggest-market/.

#### 55. **Id.**

- 56. The availability of broadband, however, may depend on the condition of the underlying electrical grid. Weather or disasters that result in power outages may also damage electronic communications systems.
- 57. The Arctic Council "is a high-level forum for political discussions on common issues to the governments of the Arctic States and its inhabitants. The Arctic Council is the only circumpolar forum for political discussions on Arctic issues, involving all the Arctic states, and with the active participation of its Indigenous Peoples." Frequently Asked Questions, Arctic Council, http://www.arctic-council.org/index. php/en/about-us/arctic-council/faq.
- 58. See The Broadband Commission, The State of Broadband in 2014: Broadband For All Annex 1 (2014), http://www.broadbandcommission org/documents/reports/bb-annualreport2014.pdf (listing broadband plans by country and the year in which they were adopted).
- 59. The degree to which these challenges manifest themselves varies across the Arctic region. For example, the lack of infrastructure in some areas may be acute, while in others roads and railways are well developed.
- 60. State of Alaska, Alaska Community Database Custom Data Queries, http://www.commerce.state.ak.us/dca/commdb/CF\_CUSTM.htm (aggregating population figures for each Alaskan city).
- 61. State Fact Sheets: Alaska, United States Department of Agriculture, Economic Research Service, http://www.ers.usda.gov/statefacts/ ak.htm
- 62. Arctic Communications Infrastructure Assessment Report, Northern Communications & Information Systems Working Group, A Matter of Survival, Arctic Communications Infrastructure in the 21st Century 77 (2011), www.aciareport.ca.
- 63. Ranajit S. Dam, Telenor Norway: Going north, takes LTE to the Arctic, Win-Win, Sept. 2011, at 33, http://www1.huawei.com/en/static/HW-094108.pdf.
- 64. Margaret Kriz Hobson, Isolated Alaska Villages Deal With Energy Problems, Arctic Deeply (Feb. 29, 2016), https://www.newsdeeply. com/arctic/articles/2016/02/29/isolated-alaska-villages-deal-withenergy-problems/
- 65. Arctic Communications Infrastructure Assessment Report, Northern Communications & Information Systems Working Group, A Matter of Survival, Arctic Communications Infrastructure in the 21st Century 12-13 (2011), www.aciareport.ca.

66. Id. at 119.

67. Fritz Bekkadal, Arctic Communication Challenges, 48 Marine Technology Society Journal 8, 10-11 (2014), https://www.researchgate. net/publication/262575866\_Arctic\_Communication\_Challenges.

#### 68 **Id**.

69. Arctic Communications Infrastructure Assessment Report, Northern Communications & Information Systems Working Group, A Matter of Survival, Arctic Communications Infrastructure in the 21st Century 10 (2011), www.aciareport.ca.

#### 70. **Id.**

- 71. Eleanor Fast. The Arctic: Transportation. Infrastructure and Communication, Parliament of Canada, Library of Parliament (Oct. 24, 2008), http://www.lop.parl.gc.ca/content/lop/researchpublications/ prb0808-e.htm.
- 72 National Telecommunications & Information Administration Response to Notice of Inquiry: Telecommunications Assessment of the Arctic Region by Heather E. Hudson 12, https://www.ntia.doc.gov/files/ntia/ heatherhudson.pdf.

73 ld.

- 74. See generally Offline and falling behind: Barriers to Internet adoption, McKinsey & Company (Sept. 2014), http://www.mckinsey.com/ industries/high-tech/our-insights/offline-and-falling-behind-barriers-tointernet-adoption.
- 75. Arctic Communications Infrastructure Assessment Report, Northern Communications & Information Systems Working Group, A Matter of Survival, Arctic Communications Infrastructure in the 21st Century 77 (2011), www.aciareport.ca.
- 76. Connecting Canadians, Government of Canada, Digital Canada 150, http://www.ic.gc.ca/eic/site/028.nsf/eng/h\_00587.html.
- 77. Government of Canada, Canadian Radio-television and Telecommunications Commission, Broadband Report Table 2.1.1 (2011), http://www.crtc.gc.ca/eng/publications/reports/broadband/ bbreport1111.htm#a2.
- 78. Government of Canada, Digital Canada 150, Nunavut and northern Quebec (Nunavik) 1 (2014), https://www.ic.gc.ca/eic/site/028.nsf/ vwapj/connecting-branche-NU-eng.pdf/\$FILE/connecting-branche-NUeng.pdf
- 79. European Commission, Digital Single Market, Broadband Coverage in Europe 2014: Mapping progress towards the coverage objectives of the Digital Agenda 73 (2014), http://ec.europa.eu/newsroom/dae/ document.cfm?action=display&doc\_id=11195
- 80. Id. at 77.

81. **Id.** 

82. Denmark, European Commission, Digital Single Market, https:// ec.europa.eu/digital-single-market/en/scoreboard/denmark#2-human capital; see also The Digital Economy & Society Index, European Commission, Digital Single Market, https://ec.europa.eu/digital-singlemarket/en/desi

83. European Commission, Digital Single Market, Broadband Coverage in Europe 2014: Mapping progress towards the coverage objectives of the Digital Agenda 83 (2014), http://ec.europa.eu/newsroom/dae/ document.cfm?action=display&doc\_id=11195.

84. Id. at 87.

85. Id.

- 86. Finland, European Commission, Digital Single Market, https:// ec.europa.eu/digital-single-market/en/scoreboard/finland
- 87. European Commission, Digital Single Market, Broadband Coverage in Europe 2014: Mapping progress towards the coverage objectives of the Digital Agenda 83 (2014), http://ec.europa.eu/newsroom/dae/ document.cfm?action=display&doc\_id=11195.

88. Id. at 112.

- 89. Id. at 108.
- 90. Iceland, European Commission, Digital Single Market, https:// ec.europa.eu/digital-single-market/en/scoreboard/iceland.
- 91. European Commission, Digital Single Market, Broadband Coverage in Europe 2014: Mapping progress towards the coverage objectives of the Digital Agenda 83 (2014), http://ec.europa.eu/newsroom/dae/ document.cfm?action=display&doc\_id=11195. More recent figures from NCOM show that, as of September 2016, 99.98 percent of Norwegian homes had access to broadband at speeds of 4 Mbit/s or greater and 78 percent had access at 100 Mbit/s.

92 Id. at 152

93 Id.

- 94. Other broadband providers in the Arctic region of Norway include: Bredbåndsfylket Troms Nornett Kysttele and Ishavslink
- 95. Trude Pettersen, Wants Broadband Capacities in the Arctic, Barents Observer (June 2, 2016), http://thebarentsobserver.com/ arctic/2016/06/wants-broadband-capacities-arctic
- 96. Norway, European Commission, Digital Single Market, https:// ec.europa.eu/digital-single-market/en/scoreboard/norway. Norwegian Communications Authority also publishes an informative annual report on the state of broadband in Norway. See Norwegian Communications Authority, Broadband in Norway 2015 (2015), http://eng.nkom.no/topical-issues/reports/\_attachment/21538?\_ ts=1520c1fb3e3.
- 97. Carlo Maria Rossotto et al., The World Bank, A Sector Assessment: Broadband in Russia 6 (2015), http://www.worldbank.org/content/ dam/Worldbank/Feature%20Story/ECA/Broadband\_in\_Russia\_Final pdf

98. **Id.** 

99. Id. at 19.

100. Akamai Technologies, Akamai's State of the Internet Q3 2015 Report 39 (2015), https://www.akamai.com/us/en/multimedia/documents/ report/g3-2015-soti-connectivity-final.pdf.



- 101. European Commission, Digital Single Market, Broadband Coverage in Europe 2014: Mapping progress towards the coverage objectives of the Digital Agenda 83 (2014), http://ec.europa.eu/newsroom/dae/ document.cfm?action=display&doc\_id=11195.
- 102 Id. at 189

103 ld

- 104. Sweden, European Commission, Digital Single Market, https:// ec.europa.eu/digital-single-market/en/scoreboard/sweden
- 105. Inquiry Concerning the Deployment of Advanced Telecommunications Capability to All Americans in a Reasonable and Timely Fashion. & Possible Steps to Accelerate Such Deployment Pursuant to Section 706 of the Telecommunications Act of 1996, as Amended by the Broadband Data Improvement Act, 31 F.C.C. 699, 735 (2016), https:// apps.fcc.gov/edocs\_public/attachmatch/FCC-16-6A1.pdf.
- 106. Id. at Appendix D.
- 107 Id. at Appendix B
- 108 **Id**.
- 109. Inquiry Concerning the Deployment of Advanced Telecommunications Capability to All Americans in a Reasonable and Timely Fashion, & Possible Steps to Accelerate Such Deployment Pursuant to Section 706 of the Telecommunications Act of 1996, as Amended by the Broadband Data Improvement Act, 31 F.C.C. 699 (2016), https://apps. fcc.gov/edocs\_public/attachmatch/FCC-16-6A1.pdf.
- 110 **Id**.
- 111. Futuristic mobile technologies foresee "IMT for 2020 and beyond," ITU towards IMT for 2020 and beyond, ITU, http://www.itu.int/en/ ITU-R/study-groups/rsg5/rwp5d/imt-2020/Pages/default.aspx.
- 112. The World Bank, Broadband Strategies Handbook 65 (Tim Kelly & Carlo Maria Rossotto eds., 2012), available at http://broadbandtoolkit. org/en/home.

113. **Id.** 

- 114. Creating a Favourable Environment for Attracting Finance and Investment in Broadband Infrastructure, a Report by the Broadband Commission Working Group on Financing and Investment, ITU, http://www.broadbandcommission.org/Documents/reports/WG-Fin-Invest-2014.pdf.
- 115. National Telecommunications & Information Administration, BROADBANDUSA: An Introduction to Effective Public-Private Partnerships for Broadband Investments 3 (January 2015). https:// www.ntia.doc.gov/files/ntia/publications/ntia\_ppp\_010515.pdf
- 116. Diana Goovaerts, Mexico Finally Launches Mobile Network Tender, Wireless Week (Jan. 29, 2016 10:26 AM), https://www.wirelessweek. com/news/2016/01/mexico-finally-launches-mobile-network-tender.

117 **Id**.

118. Juan Montes. Mexico Plans Broadband Overhaul to Boost Wireless Competition, The Wall Street Journal (Oct. 31, 2014, 6:22 PM), http://www.wsj.com/articles/mexico-plans-broadbandoverhaul-1414794178

- 119. Creating a Favourable Environment for Attracting Finance and Investment in Broadband Infrastructure, a Report by the Broadband Commission Working Group on Financing and Investment, ITU, http://www.broadbandcommission.org/Documents/reports/WG-Fin-Invest-2014.pdf.
- 120. Roslyn Layton, Municipal Broadband A Wolf in Sheep's Clothing?, TechPolicyDaily.com (Nov. 7, 2013, 6:00 AM), http://www. techpolicydaily.com/communications/municipal-broadband-wolfsheeps-clothing/

121. Id.

122 **Id** 

- 123. Creating a Favourable Environment for Attracting Finance and Investment in Broadband Infrastructure, a Report by the Broadband Commission Working Group on Financing and Investment, ITU, http://www.broadbandcommission.org/Documents/reports/WG-Fin-Invest-2014.pdf.
- 124. Id. Traditionally, sovereign wealth funds have invested in securities in major markets, but recently their investment scope has broadened to include other asset classes, include telecommunications infrastructure. ld.
- 125. Greenland calling!, Nordic Investment Bank, http://www.nib.int/news\_ publications/cases\_and\_feature\_stories/greenland\_calling.
- 126. https://www.peering-forum.eu/system/documents/5/original/ Connections\_to\_and\_within\_Iceland\_\_historical\_perspective\_3.pdf, slide 30.
- 127. Greenland calling!, Nordic Investment Bank, http://www.nib.int/news\_ publications/cases\_and\_feature\_stories/greenland\_calling.
- 128. Id. According to its website, the Nordic Investment Bank (NIB) "finances projects that improve competitiveness and the environment of the Nordic and Baltic countries. The Bank offers long-term loans and guarantees on competitive market terms to its clients in the private and public sectors. NIB is an international financial institution owned by Denmark, Estonia, Finland, Iceland, Latvia, Lithuania, Norway and Sweden. The Bank has lending operations both in and outside its member countries. NIB acquires the funds for its lending by borrowing on the international capital markets. NIB's bonds enjoy the highest possible credit rating." About NIB, Nordic Investment Bank, http://www. nib.int/about\_nib.
- 129. May 2015 Community Update, Mackenzie Valley Fibre Link, http:// mvflproject.com/construction-updates/may-2015-community-update/.

130. **Id**.

131. **Id.** 

132. **Id** 

133. ld.

134 David Thurton, Inspection reports cite environmental concerns with Mackenzie Valley fibre optic project, CBCNews (Feb. 1, 2016 9:00 AM), http://www.cbc.ca/news/canada/north/mackenzie-valley-fibreinspection-1.3428012.

135. Dave Croft, Yukon to build fibre optic line along Dempster Highway to N.W.T., CBCNews (Oct. 27, 2015 5:21 PM), http://www.cbc.ca/news/ canada/north/yukon-internet-fibre-dempster-northwestel-1.3291529.

136. **Id.** 

- 137. About, Nuvitik, http://nuvitik.ca/about/
- 138. Ivaluk, Nuvitik, http://nuvitik.ca/ivaluk/

139. **Id.** 

- 140. **Id.**
- 141. ld.
- 142. The project was begun by Arctic Fibre, a fiber optics telecommunications company. On May 18, 2016, Arctic Fibre announced that Quintillion Subsea Holdings had acquired its assets. http://www.cbc.ca/news/ canada/north/arctic-fibre-optic-internet-sale-quintillion-1.3591825.
- 143. http://www.fiercetelecom.com/press-releases/arctic-fibre-acquiredquintillion-networks.

144. **Id.** 

145. **Id.** 

- 146. Jeremy Hsu, An Internet Cable Will Soon Cross the Arctic Circle, Scientific American (June 1, 2016), http://www.scientificamerican. com/article/an-internet-cable-will-soon-cross-the-arctic-circle/.
- 147. The Broadband Initiatives Program was created as part of the American Recovery and Reinvestment Act of 2009.
- 148. Arctic states that are members of the European Union must comply with certain restrictions on tax incentives, tax credits, and other mechanisms embodied in the EU's "Guidelines for the Application of State Aid Rules in Relation to the Rapid Deployment of Broadband Networks." Communication from the Commission, EU Guidelines for the application of State aid rules in relation to the rapid deployment of broadband networks (2013/C 25/01), http://eur-lex.europa.eu/ LexUriServ/LexUriServ.do?uri=OJ:C:2013:025:0001:0026:en:PDF.
- 149. Advances in technology should continue to lower both technical and economic barriers to broadband deployment. For example, as the number of technological alternatives increases, competition can drive down costs and spur increased private-sector investment in broadband.

COVER - Spitsbergen, Svalbard, Norway | Witold Kaszkin | 123RF 1 - Nunavut, Canada | Christopher Howley | 123RF 4 - Abisko National Park, Sweden | Conny Sjostrom | 123RF 5 - Funningur, Faroe Islands, Denmark | balipadma | 123RF 8 - Barrow, Alaska, USA | Chris Arend 13-14 - Cambridge Bay, Nunavut, Canada | drferry | iStockphoto 16 - Harstad, Norway | bpfoto | 123RF 21 - Cable Innovator, Sea of Okhotsk | Global Marine Systems 26 - Cable Innovator, Svalbard, Norway | Global Marine Systems 27 - Finland | hannurama | iStockphoto 28-29 - Vlk, Iceland | fotoVoyager | iStockphoto 36 - Tromsø, Norway | Tomas Sereda | iStockphoto





# **Photo Credits**

